Author's response to reviews

Title: Previous hospital admissions and disease severity predicts the use of antipsychotic combination treatment in patients with schizophrenia

Authors:

Albert Bolstad (albert.bolstad@diakonsyk.no)
Ole A Andreassen (o.a.andreassen@medisin.uio.no)
Jan I Røssberg (j.i.rossberg@medisin.uio.no)
Ingrid Agartz (ingrid.agartz@medisin.uio.no)
Ingrid Melle (ingrid.melle@medisin.uio.no)
Lars Tanum (lars.tanum@medisin.uio.no)

Version: 4 Date: 27 July 2011

Dear Editorial Production Team

We are delighted that you are willing to accept manuscript for publication.

Formatting changes:

Major revisions

Title: Title is now identical in the manuscript and in the submission system.

Author list: The list of authors is now written in the preferred style.

Affiliations: Each affiliation is written in full, in the recommended format.

References: All references are cited within the main text and in numerical order. Former reference nr 17 is now reference nr 14.
Authors’ contributions: It is now indicated that all authors have read and approved the final manuscript.

Tables: Tables are numbered in ascending numerical order: 1, 2, 3, 4, 5 and 6. Former Table 3 a) and 3 b) are now Table 3 and 4.

Minor revisions

Corresponding author: One corresponding author is now indicated on the title page as well.

Tables: Please remove all visible vertical lines. Thank you.

Typography: We have re-checked the manuscript for typological errors. Final corrections and revisions are made in the abstract, the main text and in the reference list.

Yours sincerely

Albert Bolstad, M.D

e-mail: 'albert.bolstad@diakonsyk.no'
Department of Psychiatry Research
Diakonhjemmet Hospital
P.O. Box 85 Vinderen
N-0319 Oslo, NORWAY